Author's response to reviews

Title: Training family physicians and residents in family medicine in shared decision making to improve clinical decisions regarding the use of antibiotics for acute respiratory infections: protocol for a clustered randomized controlled trial

Authors:

France Légaré (france.legare@mfa.ulaval.ca)
Michel Labrecque (Michel.Labrecque@fmed.ulaval.ca)
Gaston Godin (Gaston.Godin@fsi.ulaval.ca)
Annie LeBlanc (LeBlanc.Annie@mayo.edu)
Claudine Laurier (claudine.laurier@umontreal.ca)
Jeremy Grimshaw (jgrimshaw@ohri.ca)
Josette Castel (Josette.Castel@fmed.ulaval.ca)
Isabelle Tremblay (Isabelle.Tremblay@fmed.ulaval.ca)
Pierre Frémont (Pierre.Fremont@crchul.ulaval.ca)
Michel Cauchon (Michel.Cauchon@fmed.ulaval.ca)
Kathleen Lemieux (Kathleen.Lemieux@ssss.gouv.qc.ca)
Caroline Rhéaume (Caroline.Rheaume@mfa.ulaval.ca)

Version: 4 Date: 5 January 2011

Author's response to reviews:

dear Editor

I added a clear statement explaining that the study was approved by the ethical acommittee (not just the consent forms. Please see: " 4) they sign the informed consent form approved by the ethics committee of the Centre de Santé et des Services Sociaux de la Vieille-Capitale. This ethics committee approved this project on June 14th 2010." on p. 9. Also, I removed the ethical approval document and funding document from the additional files.

With best regards,

France Legare